Author's response to reviews

Title: Immune reconstitution inflammatory syndrome (IRIS) associated with Acquired Immuno Deficiency Syndrome (AIDS) related gastrointestinal limited Kaposi’s sarcoma presenting as Acute intestinal obstruction: A Case report & Literature review

Authors:

Jyotirmoy Pal (dripal@rediff.com)
Ankit Shrivastav (ankit.med@gmail.com)
Dipendra Kumar Sarkar (sankit2000@gmail.com)
H S Pathak (ankit_ih@yahoo.com)
D Goswami (sankit2000@yahoo.com)

Version: 2 Date: 16 April 2010

Author's response to reviews:

Resp Sir

The corrections required have been carried out as pointed below
1) Abstract has been reformatted on the format suggested.
2) Key message has been removed.
3) Ethnicity of the patient has been mentioned.

Kindly evaluate the corrected manuscript.
I would again like to apologize for inordinate delay in response on my part.

Regards

Dr Ankit Shrivastav